Author's response to reviews

**Title:** Epidemiology of idiopathic inflammatory myopathies among commercially insured, Medicare supplemental insured, and Medicaid enrolled populations: an administrative claims analysis

**Authors:**

Karen E Tomic (karen.tomic@thomsonreuters.com)
Anthony A Amato (AAMATO@PARTNERS.ORG)
Ancilla W Fernandes (Fernandesa@medimmune.com)

**Version:** 2 **Date:** 23 December 2011

**Author's response to reviews:**

As requested by the editorial staff, we have added in text to the Methods section indicating that the databases used in this study are de-identified and fully compliant with the Health Insurance Portability and Accountability Act of 1996. Because this study did not involve the collection, use, or transmittal of individually identifiable data, Institutional Review Board review or approval was not required.